Author's response to reviews

Title: BRAF V600E mutation and KRAS codon 13 mutations predict poor survival in Chinese colorectal cancer patients

Authors:

Jing Chen (jingchen871227@126.com)
Fang Guo (wukongxiaofang@126.com)
Xin Shi (shixined@hotmail.com)
Lihua Zhang (njwaterlily@126.com)
Aifeng Zhang (aifengzh@163.com)
Hui Jin (Jinhui_hld@163.com)
Youji He (heyouji@seu.edu.cn)

Version: 3
Date: 2 July 2014

Author's response to reviews:

2nd July, 2014

Editor of BMC Cancer

Dear Editor,

Hereby we would like to resubmit to the Editorial Office of "BMC Cancer" the manuscript titled: "BRAF V600E mutation and KRAS codon 13 mutations predict poor survival in Chinese colorectal cancer patients".

We have revised our manuscript according to the request of the editor. We have stated that the written informed consent was obtained from the participants in the Methods section (page 5 lines 94-95). No other changes are made in this revision.

Thank you very much for your time and consideration, and we are looking forward to your reply.

Yours sincerely,

Prof. Youji He M.D. Ph.D